Author's response to reviews

Title: A Multi-center, Randomized, and Comparative Clinical Study of the Effect and Safety of Autologous Cultured Osteoblast Injection to Treat Fractures

Authors:

Seok-Jung Kim M.D. (peter@catholic.ac.kr)
Yong-Woon Shin M.D. (woonsos@daum.net)
Kyu-Hyun Yang M.D. (kyang@yunc.yonsei.ac.kr)
Sang-Bum Kim M.D. (sangbume@hitel.net)
Moon-Jib Yoo M.D. (miyoo@dku.edu)
Suk-Ku Han M.D. (hnsukku@catholic.ac.kr)
Soo-Ah Im M.D. (saim@catholic.ac.kr)
Yoo-Dong Won M.D. (yoodong1@catholic.ac.kr)
Yerl-Bo Sung M.D. (ybs58@sanggyepaik.ac.kr)
Taek-Soo Jeon M.D. (sirjeon@kuh.co.kr)
Cheong-Ho Chang M.D. (chc@swcellontech.com)
Jae-Deog Jang M.D. (jdjang@swcellontech.com)
Hyun-Jo Kim M.S. (angel@swcell.com)

Version: 2 Date: 20 April 2008

Author's response to reviews:

This is a clinical trial to evaluate the effect and safety of autologous cultured osteoblasts to treat fractures. To our best knowledge, this cell therapy clinical trial in orthopedic field is the first time in world.

With this successful result, new treatment options of treatment for bone disease which require bone formation in various clinical situations can be suggested.